Author’s response to reviews

Title: Hidden intra-mandibular carcinoma cuniculatum appearing in a patient with metastatic prostate cancer: A case Report

Authors:

Elyette Broly (elyette.broly@gmail.com)
Philippe Barthélémy (philippe.barthelemy@chru-strasbourg.fr)
Saït Ciftci (sciftci@strasbourg.unicancer.fr)
Christian Borel (cborel@strasbourg.unicancer.fr)
Martin Broly (martin.broly@gmail.com)
Catherine-Isabelle Gros (catherine-isabelle.gros@chru-strasbourg.fr)
Luc Marcellin (luc.marcellin@chru-strasbourg.fr)
Fabien Bornert (fabien.bornert@chru-strasbourg.fr)

Version: 3 Date: 21 Mar 2019

Author’s response to reviews:

Minor Revisions - Response letter

Matthew P. Hickey (BMC Oral Health’s editor):

Thank you for your technical comments.

All requested changes have been done. Please find them in the clean version of my manuscript.

1. Please note that the Ethics and consent for participation statement is not relevant in the case of your manuscript. Therefore, please state "Not applicable" in this section.

Changes have been done. "Not applicable" has been state in the déclaration section. (Declarations’s section, Page 13, Line 12)

2. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please amend your current Consent for publication statement so that it reflects the relevant consent information.
Changes have been done. We have added a statement about patient’s written informed consent (Declarations's section, Page 13, Line 19).

3. Please revise the Competing interests section so that it addresses all authors.

Changes have been done. "All authors declare that they have no competing interests" have been added (Declarations's section, Page 13, Line 33).

4. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

I ensure that my revised manuscript is a single clean copy.